Author's response to reviews

Title: Open-access clinical trial registries: the Italian scenario

Authors:

Paola Mosconi (mosconi@marionegri.it)
Anna Roberto (anna.roberto@marionegri.it)

Version: 4 Date: 2 October 2012

Author's response to reviews: see over
Trials

Dear Editor,

accordingly with your mail of the 1st of October, I’m sending the new version of the article: “Open-access clinical trial registries: the Italian scenario”.

As requested by the Associate Editor the text of the methods section has been moved, and it has been inserted between the background section and the results section.

Yours sincerely,

Paola Mosconi, Biol Sci D
Mario Negri Institute for Pharmacological Research